Author’s response to reviews

Title: Mitochondrial neurogastrointestinal encephalopathy as a mimic of Crohn's disease: a case report

Authors:

Ravi Patel (rp466@cam.ac.uk)
Lucia Coulter (llc32@cam.ac.uk)
Joanna Rimmer (joanna.rimmer@addenbrookes.nhs.uk)
Miles Parkes (miles.parkes@addenbrookes.nhs.uk)
Patrick Chinnery (pfc25@cam.ac.uk)
Oscar Swift (o.swift@ucl.ac.uk)

Version: 2 Date: 11 Sep 2018

Author’s response to reviews:

1. Please amend the statement in the "Consent for publication" section by providing more information, as this refers to consent obtained to publish potentially-identifying information (of patients/participants).

Information about the nature of consent now added as per requirements specified in editorial policy: "Written, informed consent for publication was obtained from patient 1 and 2." Line 131

2. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Authors names updated to full initials, within contributions section. "RP, LC, and OS all contributed to the first draft of the manuscript. RP and OS extensively revised the first draft of the manuscript and added extra information as required. JR obtained, edited, and incorporated the radiological and pathological data into the manuscript. MP and PC oversaw the project and significantly edited the manuscript.

RP and OS should be considered joint first authors of this manuscript." Lines 141 to 146

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.
All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done